University of Waterloo STAT 332 – W. H. Cherry

#2.97a

Figure 2.13. UNSTRATIFIED POPULATIONS: Calculating (approximately) a Sample Size Section headings to provide a content overview

One-Stage EPSWOR of Individual Elements

1. Precision Specified as a Probability Interval Half-width

When estimating Y−− When estimating_TY− When estimating^−P Example 2.13.1 Example 2.13.2 Example 2.13.3

2. Two Other Specifications of Precision

Example 2.13.4

3. Relative Precision

When estimating Y−− When estimating_TY− When estimating^−P Example 2.13.5 Example 2.13.6

4. Appendix: Tree Diagram Summary of Precison Specifications

Three tables

Four diagrams

Thirteen equations

2023-08-20